Author’s response to reviews

Title: Isorhamnetin 3-O-robinobioside from Nitraria retusa leaves enhance antioxidant and antigenotoxic activity in human chronic myelogenous leukemia cell line K562

Authors:

jihed boubaker (bj198024292600@yahoo.fr)
mohamed ben sghaier (medbensghaier@gmail.com)
ine skandran (ines20101979@yahoo.fr)
kamel ghedira (ghe.kamel@yahoo.fr)
leila chekir-ghedira (che_leila@yahoo.fr)

Version: 9 Date: 7 August 2012

Author’s response to reviews: see over
Editor in chief of BMC Complementary and Alternative Medicine

Dear Editor,

I have the pleasure to resubmit the revised manuscript for publication in your journal: BMC Complementary and Alternative Medicine, the revised version of our paper entitled:

« Isorhamnetin 3-O-robinobioside from Nitraria retusa leaves enhance antioxidant and antigenotoxic activity in human chronic myelogenous leukemia cell line K562» by:
Boubaker Jihed, Ben Sghaier Mohammed, Skandrani Ines, Ghedira Kamel, Chekir-Ghedira Leila.. Referenced: MS: 9718042267230860

- We tried to address the mistakes and grammatical inaccuracies as asked by the editor.
- We corrected the English language.